Author's response to reviews

Title: Absence of inferior vena cava in 14-year old boy associated with deep venous thrombosis and positive Mycoplasma pneumoniae serum antibodies - a case report.

Authors:

Boleslaw Kalicki (kalicki@wim.mil.pl)
Monika Sadecka (monika.sdc@gmail.com)
Agata Wawrzyniak (awawrzyniak@wim.mil.pl)
Piotr Kozinski (pkozinski@wim.mil.pl)
Miroslaw Dziekiewicz (dziekiewicz@wim.mil.pl)
Anna Jung (ajung@wim.mil.pl)

Version: 5
Date: 8 March 2015

Author's response to reviews:

Thank you for giving your attention to the manuscript. The authors have carefully read the e-mail, dated 24/02/2015, and, according to the suggestions and comments, implemented following changes in the manuscript:

1. The Consent section has already been included in the article. We used the recommended formula.
2. The authors completed the Competing interests section, answering questions about financial competing interests.
3. The Authors contributions section has been rewritten according to BMC standards.

We believe, that we gave a satisfying response to all the concerns. In our opinion, the article has an educational value and make a contribution to clinical practice. We look forward to receiving your comments on our efforts.